Author's response to reviews

Title: Guided online treatment in routine mental health care: an observational study on uptake, drop-out and effects

Authors:

Robin MF Kenter (r.m.f.kenter@vu.nl)
Lisanne Warmerdam (e.h.warmerdam@vul.nl)
Christine AC Brouwer-Dudokdewit (christine@debosgroep.nl)
Pim Cuijpers (p.cuijpers@vu.nl)
Annemieke van Straten (a.van.straten@vu.nl)

Version: 3 Date: 24 April 2012

Author's response to reviews:

Dear editors,

We hereby re-submit our manuscript for publication in your Journal. In response to the editors’ comment received on 2012-04-13 we would like to state that we revised our manuscript to clarify that the study was observational and that patients were offered the treatment of standard care. The previous title of the manuscript was “Guided online treatment in routine mental health care: uptake, drop-out and effects” which we changed for “Guided online treatment in routine mental health care: an observational study on uptake, drop-out and effects”. We also made it more explicit in the background section and method section that we report on data from one mental health center that decided to implement Internet treatment in standard care and to offer it to patients during the waitlist period.

We hope that the revisions we made are clear and you will consider this manuscript for publication.

Yours sincerely,

Robin Kenter, MSc
On behalf of all the authors